\documentclass[english,11pt]{article}
\setcounter{secnumdepth}{5}
\usepackage{amsmath,amssymb,epsfig,amsthm,amsfonts,psfig}
\usepackage{graphicx}
%\usepackage[T1]{fontenc}
\usepackage[ansinew]{inputenc}
\usepackage{a4}
\usepackage{babel}
%\usepackage{fancyhdr}
\usepackage{setspace}
%\usepackage{caption2}
\usepackage[flushmargin,hang]{footmisc}
%\captionstyle{hang}

% THEOREMS
\newtheorem{satz}{Satz}[section] % For books use "chapter" instead of "section"
\newtheorem{lemma}[satz]{Lemma}
\newtheorem{folgerung}[satz]{Folgerung}
\newtheorem{definition}[satz]{Definition}
\newtheorem{bemerkung}[satz]{Bemerkung}
\newtheorem{corollary}[satz]{Corollary}
\newtheorem{theorem}[satz]{Theorem}
\newtheorem{remark}[satz]{Remark}
\newtheorem{example}[satz]{Example}
% MATH
\def\eps{\varepsilon}
\def\R{\mathbb{R}}


\begin{document}


\title{\bf{Title of your thesis\vspace{1cm}}}

\author{Diploma/Semester Thesis in ... \vspace{0.25cm}\\Swiss Banking Institute\vspace{0.25cm} \\University of
Zurich \vspace{0.5cm}}

%\date{Tag.Monat 200X} %(kann spezifiziert werden, falls nicht heutiges Datum!)
\maketitle \thispagestyle{empty} \vspace{50pt}

\begin{center}

Prof.~Dr.~Thorsten Hens\\\

Dr.~Peter W\"ohrmann/Dr.~Marc Oliver Rieger/etc.\\\vspace{30pt}

Name, Surname\\99-XXX-XXX\\\

Street Nr.\\ZIP City \\phone number\\email adress

\end{center}
\begin{figure}
\begin{center}
\includegraphics[width=2.5cm]{unilogo.eps}

\end{center}
\end{figure}

\newpage
\pagenumbering{roman}

\section*{Executive Summary}




\newpage
\setcounter{tocdepth}{3} \tableofcontents

\newpage
\listoffigures \vspace{40pt}

\listoftables


\newpage
\pagenumbering{arabic} \setcounter{section}{0}

% ************************************************************************************************
\section{Introduction} % for books: use \chapter{Introduction}
%\pagenumbering{arabic} % to switch back to usual page numbers if you write a book

% ************************************************************************************************
\subsection{Subsection 1}
Bla...
\subsection{Subsection 2}
Blupp...
% ************************************************************************************************
\section[Section title]{This is the long form of a chapter title, which is to long for the headings}
% ************************************************************************************************
\section[Citations]{Examples for citations}
As in \cite{Frei:07a}, \cite{Rieger:04d} and \cite{Hens:07a}...
\section{Other examples}
\subsection{Equations}
Without numbers:
\begin{eqnarray*}
C_1(|A|^2-1)\le|\phi(A)|&\le& C_2(|A|^2+1),\\
|S(A)|&\le&C_3|A|.
\end{eqnarray*}
With numbers:
\begin{eqnarray}
u_{t}(x,t)-\mbox{\rm div}\, S(\nabla u(x,t)&=&0,\nonumber\\
u(t=0)&=&u_0.
\label{anyname}
\end{eqnarray}
See (\ref{anyname})...
\subsection{Definitions, theorems etc.}
\begin{definition}[Young measure solutions]
\label{defymparab}
We call a pair $(u,\nu)$ a {\em Young measure solution} of system (\ref{anyname}) if:
\begin{itemize}
\item $u\in H_0^1(G\times(0,T))\cap L^\infty((0,T),H_0^1(G))$, 
and $\nu:=(\nu_{x,t})$ is a family of probability measures.
\item $\langle Id,\nu_{x,t}\rangle=\nabla u(x,t)$ for a.e.\ $(x,t)\in G\times(0,T)$.
\item For all $\zeta\in H_0^1(G\times(0,T))$ we have:
\end{itemize}
$$\int_0^T\int_G\langle\phi,\nu\rangle\nabla\zeta+u_t\zeta\,dx\,dt=0.$$
\end{definition}
With this definition in hands we can state the following theorem:
\begin{theorem}[Young measure solutions for parabolic equations]
\label{t2.1}
Let $G\subset\R^n$ be a bounded domain with smooth boundary, and assume the regularity and growth
conditions for $\phi$ and $S$ stated above. Moreover assume that $u_0\in H_0^1(G)$, then there
exists a Young measure solution $(u,\nu)$ to the problem (\ref{anyname}).
\end{theorem}
You can quote the Theorem \ref{t2.1}.
\subsection{Figures}
See Fig.\ref{fig3.1}.
\begin{figure}[h]
\begin{center}
\epsfig{figure=dist.eps,width=8cm}
\caption{The function $\phi(Y):=\mbox{\rm dist}(Y,\{A,B\})^2$ with $A=\{-1\}$ and $B=\{+1\}$. -- The
cusp in $Y=0$ does not lead to a lack of differentiability for $\phi^{qc}$.}
\label{fig3.1}
\end{center}
\end{figure}
\subsection{Items}
One possibility is:
\begin{itemize}
\item[(i)]$\phi^{qc}=\phi^{**}$,
\item[(ii)]$\phi^{rc}=\phi^{**}$.
\end{itemize}
\subsection{Arrays and tables}
Here an easy example for an array:
\begin{eqnarray*}
u(1,t)&=&\left\{
\begin{array}{cl}
\frac{3^{-t}}{2}\sin(2\pi t),&t\le 2,\\
0,&t>2,
\end{array}
\right.
\\
u(0,t)&=&0.
\end{eqnarray*}
And a table:
\begin{center}
\begin{tabular}{l|lll}
&Description&$dx$&$h$\\
\hline
Experiment A&Two-well potential, initially two peaks&$0.02$&$0.02$\\
Experiment B&Sine on the boundary, stopping&$0.02$&$0.02$\\
Experiment C&Small sine on the boundary, stopping&$0.05$&$0.05$\\
\end{tabular}
\end{center}
\begin{appendix}
\section{BibTex}
Please look in the references...\\
The BibTeX-file looks as follows:\\
% The following text is printed "as if", i.e. without formating:
\begin{verbatim}
@unpublished{Frei:07a,
	Author = {Patrick Frei and Marc Oliver Rieger and Mei Wang},
	Note = {Work in progress},
	Title = {Barrier products and behavioral finance}}

@book{Hens:07a,
	Author = {Thorsten Hens and Marc Oliver Rieger},
	Date-Added = {2006-07-02 16:35:32 +0200},
	Date-Modified = {2006-07-02 16:44:34 +0200},
	Publisher = {Springer Verlag},
	Title = {Financial Economics -- A Concise Introduction to Classical and Behavior Finance},
	Year = {scheduled for 2007}}

@article{Rieger:04d,
	Annote = {Marc Oliver Rieger},
	Author = {Marc Oliver Rieger and Mei Wang},
	Journal = {Economic Theory},
	Pages = {665--679},
	Title = {Cumulative {P}rospect {T}heory and the {S}t.~{P}etersburg Paradox},
	Volume = {28},
	Year = {2006}}
\end{verbatim}
It is automatically used to build the bibliography out of the references made in the text.
\end{appendix}
\bibliographystyle{alpha} 
%\addcontentsline{toc}{chapter}{Bibliography}
\bibliography{testbib.bib} % Position of your .bib file
\end{document}
%
% For Texing use once LATEX, once BIBTEX and twice LATEX!!!
%
\end{document}
